Author’s response to reviews

Title: High-viscosity bone cement for vertebral compression fractures: a prospective study on intravertebral diffusion and leakage of bone cement

Authors:

meiyong wang (wanglei10008@sina.com)

qunhua jin (11083861@qq.com)

Version: 4 Date: 05 Aug 2020

Author’s response to reviews:

All data generated or analysed during this study are included in this published article [and its supplementary information files]. Authors' names have been changed into full initials, in the Authors’ Contributions section.